Author’s response to reviews

Title: Carbonic anhydrase IX in oligodendrogial brain tumors

Authors:

Sally Jarvela (sally.jarvela@sci.fi)
Seppo Parkkila (seppo.parkkila@uta.fi)
Helena Bragge (helena.bragge@pshp.fi)
Marketta Kahkonen (marketta.kahkonen@pshp.fi)
Anna-Kaisa Parkkila (annakaisa.parkkila@pshp.fi)
Ylermi Soini (ylermi.soini@ppshp.fi)
Silvia Pastorekova (seppo.parkkila@uta.fi)
Jaromir Pastorek (seppo.parkkila@uta.fi)
Hannu Haapasalo (hannu.haapasalo@pshp.fi)

Version: 4 Date: 13 December 2007

Author’s response to reviews:

December 13th 2007

Dear Editor,

On behalf of all the authors I thank you for accepting our manuscript for the publication. I have now made the requested changes to the format of our manuscript and below you will find these changes listed point-by-point.

With all the best regards, Sally Järvelä, M.D.

Changes made
1. Title, sentence case  page 1, textline 1. Carbonic anhydrase IX in oligodendrogial brain tumors.
2. Affiliation 2, city and country added  page 1, textline 8. Institute of Medical Technology, University of Tampere and Tampere University Hospital, Tampere, Finland.
5. Section headings, sentence case. All the section headings are now corrected.
6. Abstract, structure  page 2. The structure is now corrected, please see page 2.
7. Introduction, rename  page 3. This section is now renamed as Background.
8. Materials and methods, rename  page 6. This section is now renamed as Methods.
9. Conclusions  page 19. The final paragraph is now placed under heading Conclusions.
10. The end of the manuscript, rearrange page 20. The rearrangements have been done and the order is now as requested: Conclusions, Competing interests, Authors contributions, Acknowledgements, References.

11. The additional file with contributions. Deleted.